Daily status 29 October

10/29/13
Summary:

® Partial tear-down starting today
¢ Still breezy -- 15m/s winds

Actions past 24 hours:
® Continued Qsoil.base plot testing
To dos:
® Tear-down of near/far rad/soil
Sensor status (abbreviated):
QCtables/cockpit show all signals coming in.
P: noticed that NNE deviates from others in the crater -- presumably due to the strong winds
SPN1: Delta-T has suggested some tests, but won't have time to pursue.
Riw.in: warm event last night at all sites -- must have been a low cloud

kh2o.flr: sharp lower kh2oV event happening right now <possibly a bit of rain, but hard to believe as | overlook the crater>
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